# CARE WORK AND THE ECONOMY

Advancing policy solutions with gender-aware macroeconomic models

## QUALITY OF CARE AND WORKING CONDITIONS: UNDERSTANDING THE CARE WORKERS' PERSPECTIVE

Shirin Arslan, American University Arnob Alam, American University Maria Floro, American University Seung-Eun Cha, University of Suwon Eunhye Kang, Seoul National University

CWE-GAM Working Paper Series: 22-02 Program on Gender Analysis in Economics (PGAE) American University, Washington, DC DOI: <u>https://doi.org/10.17606/xb8d-hv82</u> May 2022

This working paper is part of the Care Work and the Economy (CWE-GAM) Project, based at American University's Program on Gender Analysis in Economics (PGAE). The authors would like to thank the participants of the virtual 2021 CWE-GAM Project Annual Meeting for their helpful comments and Thomas Hungerford and Amos Golan for providing valuable assistance. The authors would like to acknowledge the financial support of the William and Flora Hewlett Foundation for this report. The findings, views, and conclusions or recommendations expressed in this material are strictly those of the authors and do not necessarily reflect those of the William and Flora Hewlett Foundation.

Corresponding Author: Maria Floro, Department of Economics, American University, Washington DC, USA. <u>mfloro@american.edu</u>

#### ABSTRACT

Korea's puzzling situation of continued heavy reliance on family care for older persons and young children despite the growing supply of paid care services and workers has brought more attention to concerns about the quality of paid care. While several factors influence the performance and delivery of care provided by workers, this paper focuses on the care worker's sense of responsibility for the well-being of the care recipient and its relationship with working conditions. Care work is in large part, emotional labor and therefore has motivational dimensions that have serious consequences on the recipient's well-being. A worker's commitment to the care recipient's well-being is not static, however. It can change, depending on circumstances such as working conditions including job quality and the intensity of care work. In this sense, the latter can affect the quality of care. We conduct tobit and generalized maximum entropy (GME) analyses using the 2018 Care Work and the Economy (CWE-GAM) Korean childcare and eldercare worker survey data to assess the relationship between commitment levels of care workers and their working conditions. Our results suggest that a lack of adequate training and longer commute times, particularly for childcare workers, are associated with lower levels of commitment. Our results also suggest that having stable work schedule and ease in dealing with family members, particularly for eldercare workers, are correlated with higher levels of commitment.

Keywords: quality care services, childcare, eldercare, paid caregivers, job quality, South Korea

JEL codes: J13, J14, J81, D91, J28, J490

#### Quality of Care and Working Conditions: Understanding the Care Workers' Perspective<sup>1</sup>

#### 1. Introduction

Korea currently faces a puzzling situation that is also emerging in other high and middleincome countries: continued heavy reliance on family care for the frail older persons and young children despite the ample supply of paid care services (Cha et al., 2022). Over the last two decades, the Korean government has increasingly invested in care services including subsidies for childcare and early childhood education (ECE), and long-term care (LTC) insurance for eldercare. While the number of paid childcare and eldercare workers in Korea has increased in the last decade or so, a significant amount of unpaid care work continues to be performed by family members (Kang et al., 2021).

Korea's adults (aged 18 and older) provide an estimated 87 million hours of unpaid care per day. This roughly translates into 11 million full-time workers (assuming 8-hour shifts) providing paid replacement for the unpaid care time (Suh, 2021).<sup>2</sup> Much of the burden of family care falls heavily on female members, who, on average, spend more than 50 hours a week (Cha et al., 2022).

One reason for this conundrum has to do with the affordability as well as variability in the quality of paid care services, making it difficult for families to substitute paid services for family caregiving. In the case of childcare, studies in Korea show that the lack of affordable, quality care has compelled even dual-earning households to heavily rely on informal care provided by family members (Kim and Jeong, 2006; Sung, 2017). Similar reasons are made by older person caregivers particularly daughters and daughters-in-law who cite serious concerns regarding quality of care as reasons for performing eldercare themselves (Choi and Kim, 2013; Lee, 2018; Song, 2014).

Discussions about quality care were already under way in Korea even before the COVID-19 pandemic, with serious concerns around child- and elder- abuse by care-workers and day-care teachers, *yoyangbohosa, etc.*).<sup>3</sup> In some childcare centers and eldercare facilities, monitoring has been put in place using surveillance cameras.<sup>4</sup> Policy experts have pointed out that care workers'

<sup>&</sup>lt;sup>1</sup> The authors would like to thank Jooyeoun Suh, Amos Golan, and the participants of the 2021 International Association for Feminist Economics Conference for their helpful comments.

<sup>&</sup>lt;sup>2</sup> On average, women in Korea performed 29.2 hours a month of unpaid care, while men performed 13.4 hours (Suh, 2021).

<sup>&</sup>lt;sup>3</sup> The term *yoyangbohosa* is a newly defined job category in South Korea that refers to certified care workers in both homes and institutions.

<sup>&</sup>lt;sup>4</sup> Training programs for care workers in several care facilities have put emphasis on care ethics to help reduce incidence of abuse.

abusive behavior is related to care worker's stress level (Kim, 2020), how they value their work (Kim et al., 2021; Oh and Lee, 2021) and the lack of social support (Lee and Kim, 2018).

The issue of quality care is admittedly complex and yet it is critical in understanding the care conundrum in Korea. Care provisioning is distinct from other types of services in that it requires personal attention, is usually provided on a face-to-face basis, and often for persons needing assistance in performing daily activities and bodily functions (Waerness, 1987). This distinction makes the paid care sector particularly susceptible to quality problems (Folbre, 2006). Examining countries that witnessed a rapid expansion of the paid care sector, Meagher (2007) raises a pertinent question: "How are good caring relationships sustained between strangers who are systematically thrown into intimate contact with each other?" (p. 34). While some argue that paid care workers may be unlikely to provide the same quality of care and emotional support that a loving family member or kin can offer (Moon and Cha, 2020), others have pointed out that because of the specialized training they receive and the opportunity cost of family caregiver's time, paid care workers are more effective in providing care and thus may provide higher quality of care (Banuriet al., 2019). Moreover, care work, regardless of whether they are paid or unpaid, has moral/motivational dimensions, as well as practical dimensions. Commitment or a strong sense of responsibility in the delivery of care services therefore plays a critical role in determining the quality of care.

A care worker's level of commitment to the recipient's well-being is not static; it can change depending on circumstances as well as the work environment or setting. While it is likely that a care worker's sense of responsibility for the recipient increases as more time is spent together, it can also decline because of poor stressful working conditions. For example, stressful working conditions such as difficulty in dealing with recipient's family members, long commute hours as well low job quality in terms of lack of stable/predictable work hours, irregular work contract, and inadequate training can adversely affect a worker's level of commitment. In this paper, we examine the extent to which this commitment level is associated with one's working conditions such as job security, adequacy of training, care work intensity, and nature of the relationship with recipient's family. It fills a gap in the literature by examining a critical ingredient – care worker's sense of responsibility – in the delivery of care services, which has thus far received little attention in studies assessing the quality of care.

We focus on childcare and eldercare workers in South Korea in various settings such as private and public institutions and recipients' homes. We examine the relationship between care workers' expressed level of commitment towards the recipient's well-being and working conditions using the childcare and eldercare worker survey collected in 2018 by Gallup Korea among 600 workers in Korea, as part of the Care Work and the Economy Project. Given the small sample, we conduct both Tobit and general maximum entropy (GME) tests. We create sampling weights to address any selection bias that may have occurred in the sampling process. These weights use selected demographic characteristics including the type of care arrangement and geographic region to ensure that the metrics derived from the sample data are representative of the population of childcare workers and older person care workers in Korea.

This paper proceeds as follows: Section 2 discusses the context while Section 3 provides the analytical framework. Section 4 provides a description of the data, the proxy indicators used in our analysis, and methodology. Section 5 presents the Tobit and GME estimates and discusses the results. Section 6 addresses potential selection bias and endogeneity issues. Section 7 concludes with policy implications.

#### 2. Background and Context

Over the last decade, Korea's population of elderly aged 80 + more than doubled, from 0.86 million (in 2009) to 1.8 million (in 2019). By the year 2060, it is projected that the number of persons aged 65 or over will exceed 80% of the working-age population. Alongside Korea's rapid population aging, fertility rate has consistently declined since 1960, reaching a record low of 0.84 in 2020.<sup>5</sup> The continued decline in fertility rate (significantly below replacement levels) and the rapid growth of elderly have raised grave economic and social concerns with regards to future labor supply, pensions, economic growth and social reproduction. At the same time, new needs are emerging with rising standard of living such as greater expectations for quality care services. In the case of childcare for example, many parents have come to expect "enriched and educationally focused services to be offered" (Meagher 2007, p. 34). Likewise, expectations for quality and affordable eldercare have also changed. According to the 2002-2018 national social

<sup>&</sup>lt;sup>5</sup> In 2020, South Korea's population declined for the first time, with the number of births down 10 percent from 2019 (Lee, 2021).

statistics survey, only 27% of Koreans agreed that the family is responsible for caring of a frail/elderly and the expectation of the government's role in care has increased dramatically in recent years (Kim, 2019).

In response, the Korean government has been expanding public investment in care provisioning by implementing a universal childcare service policy, broadening the scope of LTCI (Long-term Care Insurance) coverage to include dementia patients and those without severe functional limitations, as well as strengthening the Community Care system by local governments (See Appendix A). Consequently, the paid care workforce expanded drastically. While the population of children under 5 continued to shrink over the last decade, the childcare workforce grew from 207,000 childcare workers in 2009 to more than 331,000 in 2019 (Ministry of Health and Welfare, 2020). The eldercare workforce also doubled during this period, from 140,000 in 2009 to 287,071 in 2018.

Despite the expansion of government support and the rapid growth of the private care sector, the care burden on household members has yet to improve (Cha et al., 2021). Family caregivers continue to view caregiving as a burden and experience significant opportunity cost (Moon and Cha, 2020). As in other countries, the primary caregivers in Korea are typically women, who perform a large share of the total care work, even with the utilization of paid care services (Choi, 2009; Kim, 2010; Choi et al., 2012; Song, 2014; Choi et al., 2014; Lee et al., 2015; Song, 2016; Chung et al., 2017; Cha et al., 2022). Before the onset of the COVID-19 pandemic, about 48.3% of total childcare in South Korea were provided by family members (KICCE, 2018). More than a third of Korean women in their 30's and 40's report experiencing the double burden of care i.e., taking care of both children and frail elderly parent(s) (Song, 2014).

The unequal burden of unpaid care on women in Korea has held the country back from achieving gender equality. Women returning from career breaks from childbirth or childcare often re-enter the labor market as non-regular workers with low-paying jobs, which is a key factor behind the Korea's large gender wage gap (31.5% in 2020), compared to the OECD average of 12.5% (OECD, 2019)<sup>6</sup>. At the same time, Korea's overall employment rate remains below the OECD average, largely due to low female employment, with the fourth largest gender employment gap (18 percentage points) among OECD countries (OECD, 2020). Cultural practices and social norms such as a work culture involving long hours, filial piety, and socially ascribed roles for mothers,

<sup>&</sup>lt;sup>6</sup> OECD, 2019, Gender wage gap (indicator). doi: 10.1787/7cee77aa-en (Accessed on 24 January 2022)

daughters, and daughters-in-law that continue to persist, as well as pro-natalist policies that largely involve financial support for families in Korea have had limited impact so far (Jones, 2019).

#### 3. Nature of Care Work and the Quality of Care

Our study builds upon the existing work on the nature of care work and emotional labor. We also review the literature on assessments of the quality of paid care services. Care work – whether nursing, eldercare, or childcare – is essential for human life, the creation of capacities, and the continuation of social relations and economics processes. Care work is unique in that it is a form of emotional labor that encompasses human relations skills, communication skills, emotional effort, and responsibility for client well-being (England and Farkas, 1986; Steinberg, 1999). This emotional component of care work requires that 'the person doing the caring is inseparable from the care given" (Himmelweit, 1995 p. 8) and that carers need to *care about* and *care for* those they tend to as the latter's well-being depends upon this service (Folbre, 2006; Nelson, 2010; Hong and Seo, 2012). The critical role of the care worker's sense of responsibility for the well-being of the recipient in delivery quality care services can be observed in healthcare and elderly care facilities (Teng et al., 2009; Kilaberia, 2020), as well as other types of services where competition is based not only on prices but also on quality and therefore, incentives for quality provision are warranted (e.g., education services).

As feminist economists have pointed out, the performance and delivery of care by paid workers are not purely motivated by self-interest or monetary reward. Intrinsic motivation and a sense of responsibility for the well-being of the recipient play a critical role in determining the level of effort provided by care workers (Tronto, 1987; Folbre and Weisskopf, 1998; Nelson, 1999; England, 2005; Meagher, 2007; Himmelweit and Land, 2010). However, the paid care sector is also particularly susceptible to quality issues for several reasons: those receiving care often lack the ability, information and/or experience required to assess the quality of what they are receiving; both inputs and outputs of care are difficult to measure; and while the intrinsic motivation among workers helps ensure quality, workers may have little control over their work environment, thus affecting the manner in which the service is delivered (Folbre, 2006).<sup>7</sup>

<sup>&</sup>lt;sup>7</sup> As noted by Moon and Cha (2020), care workers often have little or no control of the care situation. They often are not allowed to decide on what to do or what should be done in critical moments.

As with other essential services, quality care services also require a solid *provider – recipient* relationship, grounded in firm trust between the care recipient (or the recipient's family) and the care provider. However, in the case of child care and elder care services, establishing trust in provider – recipient relationships is difficult, especially in cases where care services are customized to the recipient rather than standardized (Folbre, 2008). The difficulty in establishing and sustaining good care relationships between "strangers who are systematically thrown into intimate contracts with each other" can be observed in countries that experienced a rapid growth of the paid care sector (Meager, 2007).

The quality of care delivered by paid care workers depends on many factors. Some are obvious and easy to observe, such as adequate training (Bruchinal et al., 2002; Burke et al., 2006; Nicol et al., 2005), organizational culture and commitment to safety (Gershon et al., 1995; Kwon and Hong, 2017), as well as working conditions and structural factors such recipient-to-worker ratio, worker's autonomy, wages and benefits, among others (Landsburgis, 1988; Felton, 1998; Yassi et al., 1991; Burchinal et al., 2002; Burchinal, 2018; Ruffini, 2021). Other are less obvious and difficult to measure, such as the relationship between care worker and recipient (Shin and Hyun, 2015), care worker's intrinsic motivation and the level of trust between recipient and provider (Banuri, 2019).

Assessing the quality of care services involves the crucial tasks of gathering information regarding the workers' qualifications as well as monitoring and supervision. To monitor and assess the quality of care services within facilities, a plethora of indices, scales and assessment instruments are available.<sup>8</sup> These instruments are typically based on measurable attributes and quantifiable procedures or outcomes, and they are often specific to the type of care being provided. Efforts to assess the quality of care services typically rely on such standardized metrics and assessment instruments, focusing on various aspects of care provisioning, such as the performance of care activities and interactions within facilities (Mashburn et al., 2008; Rubio-Codina et al., 2019), outcome of care services such as follow-up health assessments and treatment outcomes (Rios-Zertuche et al., 2019; Kondo, 2015), level of satisfaction by the care recipient or care recipient's guardians (Shin and Hyun, 2015), and measurable attributes of the care workers such as their education, training received and working conditions (Burchinal et al., 2002; Kwon and

<sup>&</sup>lt;sup>8</sup> Examples include Family Day Care Rating Scale (FDCRS), Early Childhood Education Ratings Scale (ECERS), Caregivers Interaction Scale (CIS), to name just a few (Harding, 2002; Colwell et al., 2013; Early et al., 2018; Westerberg et al., 2018).

Hong, 2017; Burchinal, 2018). However, as Joling et al. (2018) finds through a systematic review of hundreds of quality indicators, indicators measuring care outcomes and non-clinical aspects are relatively scarce and most indicators do not meet standards of high methodological quality.

While the intrinsic motivation of the care worker is a critical ingredient in quality care provisioning, the preceding literature review shows that it has received little attention in efforts to assess the quality of care-services. The care worker's level of commitment is not only crucial in the performance of basic tasks such as dressing, feeding, bathing, giving medication, etc. It is also critical in building personal connection and meeting the emotional and developmental needs of the recipient through activities such as reading to recipient or listening to their story and providing comfort when the recipient is feeling troubled. As Susan Eaton (2005) puts it, these are activities or tasks that make all the difference if you're living in a nursing home or spending the day in a care center.

A worker's sense of responsibility for (or commitment towards) their recipient's safety and wellbeing is not static, however. It is influenced by a myriad of factors, many of which are not easily measured such as the nature of the relationship with the recipient, and certain worker's characteristics including level of empathy, patience, and conscientiousness. A worker's commitment level can also be affected by their working conditions e.g., intensity of caregiving, relationship with family members, commute hours, type of work contract, having regular breaks, receiving adequate training, etc. In this paper, we argue that working conditions and job quality are interrelated with this commitment level, and therefore the quality of care. For instance, a poor work environment can undermine a care worker's positive attitude and motivation. Stressful working conditions or low job quality in terms of lack of stable/predictable work hours, high number of care recipients at any given work period, job insecurity and the absence of benefits that facilitates a worker's healthy work-life balance can adversely affect a person's level of commitment. This may lead not only to high turnover rates but also to lower quality of care provided. Other factors that have to do with the geographical setting such as length of commute time can also affect a worker's commitment level. Heavily urbanized countries like South Korea have witnessed the lengthening of average commute times especially in metropolitan areas such as Seoul and Busan. With the removal of monthly travel allowance by some care institutions and employment agencies, many workers are forced to rely on public transport (buses and subways).

which considerably lengthens their commute time compared to a hybrid mode of travel involving short taxi rides and subways.

Additionally, there is a connection between the worker's level of commitment and intensity of care work as determined by the need of the recipient for constant attention or continual assistance to perform basic functions. In this case, the relationship is likely to be bi-directional in the on one hand, the intensity of care work can strengthen the attachment and emotional bond with the carer, which help maintain and even heighten the latter's sense of commitment. It may also be the case that care workers with strong sense of commitment are more likely to take on care jobs that are more intense such as caring for very young children or frail elderly who are immobile or with severe dementia. Nonetheless, persistent intensity of caregiving can lead to exhaustion, tiredness, or even burnt-out, which can eventually sour the worker's level of commitment.<sup>9</sup> Therefore, the intensity of care work may or may not be associated with lower levels of commitment.

Our study empirically examines the association between the expressed sense of responsibility of care workers and their working conditions, including job quality, relationship with (family of) recipient, commute time, and the intensity of care work. We predict that a worker's level of responsibility and commitment towards the recipient's well-being correlates with the realities of the work environment including job quality, relationship with (family of) recipient, adequacy of training received, care work intensity, and other factors. In other words, poor working conditions i.e., unpredictable work schedule, long commute, inadequate or lack of appropriate training, job insecurity, etc. can undermine the worker's morale and level of commitment to the care recipient. If, as we have argued in this section, a care worker's sense of responsibility affects the quality of care, then the latter may suffer as a worker's morale and commitment level are adversely affected by the working conditions.

#### 4. Empirical Analysis

#### 4.1 Data and Methodology

Our analysis uses the 2018 Care Work and the Economy survey data collected by Gallup Korea. The purposive sample consists of 300 eldercare workers and 300 childcare workers in public and private care institutions across South Korea, including Seoul/Metropolitan Area (Seoul,

<sup>&</sup>lt;sup>9</sup> A few studies have shown that work intensity in terms of difficult physical and emotional labor entailed is associated with lower quality of care (Kim et al., 2018; Kim and Yeom, 2016).

Incheon, Gyeonggi-do, Gangwon-do), Chungcheong Area (Daejeon, Sejong, Chungbuk, Chungnam), Honam Area (Gwangju, Jeonbuk, Jeonnam), Gyeongbuk Area (Daegu, Gyeongbuk), Gyeongnam Area (Busan, Ulsan, Gyeongnam). The sampling design of childcare and eldercare workers took into account the stratification by geographical region and occupational categories (institutional worker, home-based worker, or informal worker) (Jun et al., 2021).<sup>10</sup> Inverse sampling probability weights were obtained using data from the 2017 Day Care Centre Statistics Yearbook (National Statistics Office, 2017) and 2017 Long-Term Care Insurance Statistical Yearbook (National Health Insurance Corporation, 2017) to adjust the sample distribution and make it representative of the childcare and eldercare worker population (Suh, 2020).<sup>11</sup> Appendix B describes the methodology for constructing the sampling weights.

The survey question "How much responsibility do you feel for the health and safety of your care recipient(s)?" is used as our measure of expressed commitment or sense of responsibility by the care worker. An important caveat regarding the main variable of interest is that it is based on the respondent's own assessment. Moreover, the data is cross-sectional and hence we are unable to evaluate the direction of change over time. Finally, the responses are bounded between 0% (not my responsibility at all) and 100% (entirely my responsibility). Since responses are bounded, we examine the relationship between the worker's level of commitment and proxy indicators of their working conditions using Tobit regression analysis. We also conduct an entropy econometrics regression analysis to test the robustness of the results.

Table 1 describes the characteristics and working conditions of the care workers in our sample. Reflecting the dominance of women in Korea's paid care sector, most (95 percent) of the care workers in our sample are women, their weighted mean age is 52.5 with elder care workers being older on average (54.4 years) compared to childcare workers (47.3 years). Most of the care workers in our sample completed high school (72%), lived with a spouse (85%), and in dual-earning households (77%). Majority of them also work in a metropolitan area (73%) and about half (50.4%) are regular or contract employee with a signed contract.

<sup>&</sup>lt;sup>10</sup> Eldercare workers in institutional facilities work in nursing homes and daycare centers, excluding hospitals. Home-based eldercare workers work in the older person's own home and are funded by National LTCI, while informal eldercare workers are hired by families or the elderly without written or formal contracts e.g., lived-in carers. Institutional childcare workers are employed in public, private or corporate daycare centers. Homebased childcare workers are hired through agencies while informal childcare workers are hired by families without formal contracts, e.g., informal babysitters.

<sup>&</sup>lt;sup>11</sup> For informal workers, regional informal worker population was estimated using informal sector share of GDP. See Appendix B for details.

	All Workers	Childcare Workers	Eldercare Workers
A. Worker Characteristics			
Average Age (years)	52.5	47.3	54.4
Care Work Experience (mean, in years)	4.7	5.62	4.4
Gender (% distribution)			
Female	94.8%	95.0%	94.8%
Education (% distribution)			
No schooling	0.1%	0.0%	0.2%
Primary	1.9%	1.5%	2.0%
Middle School	6.3%	0.5%	8.4%
High School	71.8%	56.6%	77.3%
College	19.3%	40.1%	11.8%
Graduate	0.6%	1.3%	0.4%
Number of care work licenses (% distribution)			
0	16.2%	35.4%	9.3%
1	68.0%	46.0%	76.0%
2	12.1%	14.0%	11.4%
3+	3.6%	4.6%	3.3%
Has a Spouse (% distribution)			
Yes	85.3%	87.0%	84.7%
Dual-Earner Household (% distribution)			
Yes	77.2%	83.2%	75.1%
B. Working Conditions			
Number of care recipients (mean) <sup>1</sup>	2.7	2.3	2.9
Work hours (mean) <sup>2</sup>	39.4	37.3	40.2

# Table 1 Characteristics and Working Conditions of Care Workers, by Type of Worker

Average commuting time to work (mean in minutes) <sup>3</sup>	46.2	41.3	48
Need to watch care recipient at all times $(\% \text{ distribution})^4$			
Yes	49.0%	68.9%	41.7%
Extra work hours (% distribution) <sup>5</sup>			
Yes	26.7%	36.6%	23.2%
Metro (% distribution) <sup>6</sup>			
Yes	73.3%	78.3%	71.5%
Care work is physically difficult (% distribution) <sup>7</sup>			
Yes	65.7%	57.6%	68.7%
Has a stable work schedule (% distribution) <sup>8</sup>			
Yes	61.2%	61.5%	61.1%
Has regular holiday leaves (% distribution) <sup>9</sup>			
Yes	80.6%	83.2%	79.7%
Family is relatively easy to deal with (% distribution) $^{10}$			
Yes	28.9%	37.5%	25.4%
Regular or contractual employee with a signed contract (% distribution) <sup>11</sup>			
Yes	49.6%	38.2%	53.7%
Institution-based worker (% distribution) <sup>12</sup>			
Yes	51.6%	32.7%	58.4%

*Note*: Calculated using the 2018 CWE-GAM Korean Childcare and Eldercare Workers Survey data. Statistics based on respondent's answer to the following survey questions:

1. How many care recipients have you taken care of over the past week?

2. Over the past month, how many hours per day did you do care work on average? (Sum of weekday and weekend hours)

3. How much time does it take to commute to work from your home on average?

4. I need to watch my care recipient at all times (agree/strongly agree =1, yes)

5. I work more hours than the standard number of hours (agree/strongly agree=1, yes)

6. Opening survey question completed by survey investigator on the location of care work provided.

7. In general, how much physical difficulty do you have taking care of the child or elderly person? (Slightly/very difficult=1, yes)

8. There are times when my work schedule gets cancelled without notice (strongly /somewhat disagree=1, yes)

9. I can apply for holidays when I want to (strongly/somewhat agree=1, yes)

10. It is very difficult to deal with the care recipient's family members (strongly/somewhat disagree=1, yes)

11. what type of employment do you have at your current workplace (regular employee or contract up to 2 years), and have you signed an official written labor contract related to your current care work (yes or don't know)?

12. Main workplace (Work at an institution or care center)

Table 1 also shows the educational level of the sampled care workers: about 12% of eldercare workers hold a college degree, compared to 40% of childcare workers. Most care workers hold at least one professional license; 35 percent and 9.3 percent of childcare and elder care workers respectively work without any license. The mean years of care work experience is higher for childcare workers (5.6 years) compared to elder care workers (4.4 years). Majority of elder care workers (58.4%) are institution-based workers, whereas childcare workers are more likely to work in care recipient's home.

On average, the care worker respondents in our sample cared for 2-3 recipients, worked about 40 hours a week, and spent roughly 46 minutes on commute each day. Compared to childcare workers, eldercare workers on average, tend to care for more recipients, work more hours per week, and spend longer time commuting to work. About a quarter (26%) of the sample reported that they work more hours than stated in original employment agreement. Nearly 40% reported having an unstable work schedule. More than half of the paid care workers don't have a signed labor contract or regular employment; this is the case for 62.5% and 74.7% of childcare and eldercare workers, respectively. A large majority (71%) also reported difficulty in dealing with care recipients' family members, particularly among eldercare workers (74.7%) compared to (62.5%) childcare workers.

In terms of job quality, about 61% of workers reported to have a stable work schedule, 81% have access to regular holidays, and nearly half have secure employment (defined as a regular employee or a contract employee with a signed contract). Less than one-third of respondents agreed that it is relatively easy to deal with recipient's family, which we use as a proxy for relationship with family of recipient.

In terms of care work intensity, about half (49%) of the paid care workers reported that their care work requires them to "watch recipient(s) at all times" during working hours; this is more pronounced among childcare workers (68.9%) than among eldercare workers (41.7%). More than a quarter of employees responded that they worked extra hours than the standard (more than a third for childcare workers), nearly two in three responded that care work is physically difficult (69% of eldercare workers). This indicates that care work is intense/difficult for a large segment of the care workers and there are significant differences in the dimension of difficulty between eldercare and childcare.

The frequency and cumulative distributions of our main variable of interest i.e., level of expressed commitment or sense of responsibility of the eldercare and childcare workers are given in Figures 1 and 2 respectively and ranges from 0% (not my responsibility at all) to 100% (entirely my responsibility). Overall, the mean percentage level of responsibility reported by the respondents is 71.6%. Childcare workers tend to report higher level of responsibility (79.5% on average), compared to elder care workers (68.7% on average) as shown in Table 2.

<b>Responsibility for Care Recipient</b>	Mean	Std. Dev
Type of Worker (in percent)		
Table 2. Average Care Workers' Rep	orted Level of Responsik	bility for Care Recipient, by

<b>Responsibility for Care Recipient</b>	Mean	Std. Dev
All Workers	71.60%	20.60%
Childcare Workers	79.50%	17.30%
Eldercare Workers	68.70%	21.00%

*Note*: Statistics are based on 2018 CWE-GAM Korean Childcare and Eldercare Worker Survey respondent's answer to the following question: *"How much responsibility do you feel for the health and safety of your care recipient(s)?"* The responses ranged between 0% (not my responsibility at all) and 100% (entirely my responsibility).

### Figure 1. Frequency Distribution of Care Worker's Reported Level of Responsibility for Well-being of Recipient, by Type of Care Worker



Data Source: 2018 South Korea Paid Careworker Survey- Childcare. Care Work and the Economy Project Field Work Data, Center for Transnational Migration and Social Inclusion, Seoul National University; and 2018 South Korea Paid Careworker Survey- Eldercare. Care Work and the Economy Project Field Work Data, Center for Transnational Migration and Social Inclusion, Seoul National University

# Figure 2. Cumulative Distribution of Care Workers' Reported Level of Responsibility for Well-being of Recipient, by Type of Care Worker



**Data Source**: 2018 South Korea Paid Careworker Survey- Childcare. Care Work and the Economy Project Field Work Data, Center for Transnational Migration and Social Inclusion, Seoul National University; and 2018 South Korea Paid Careworker Survey- Eldercare. Care Work and the Economy Project Field Work Data, Center for Transnational Migration and Social Inclusion, Seoul National University

#### 4.2. Methodology

An underlying argument of this paper is that quality of care services is related to the working conditions of workers, and an important dimension of this relationship has to do with the latter's effect on the workers' commitment level. In this section, we empirically examine the following hypothesis: are better working conditions in terms of job quality and care work intensity associated with higher level of expressed commitment towards the well-being of the recipient? We use the following indicators for job quality and care work intensity: a) stable work schedule, b) job security, which is proxied by a dummy variable indicating regular employment status or having either a labor contract for up to 2 years or a signed written agreement, c) if worker lacks adequate training, d) whether recipient needs to be watched at all times, e) pleasant working environment proxied by a dummy variable indicating the ease in dealing with care recipient's family, f) number of recipients currently being cared for, and g) regular occurrence of working extra hours. In addition, we consider care worker's average commute time as part of the working conditions,

based on the Korean care workers' concern regarding long commute to and from their place of work .<sup>12</sup>

Since the dependent variable is bounded, we use censored regression models to test our hypothesis. For the Tobit model, we assume that the observed dependent variable,

$$y^{t} = \begin{cases} 0 \text{ if } y^{t*} < 0\\ y^{t*} \text{ if } 0 < y^{t*} \le 1\\ 1 \text{ if } y^{t*} > 1 \end{cases}$$

That is, our observed values  $y^t$  are bounded between 0 and 1 for the underlying latent variable  $y^{t*}$ . We then estimate the model using a maximum likelihood (ML) approach.

Given the small sample size, we also conduct an entropy-based econometric analysis. This method is deemed appropriate because it does not require a restricted assumption on the distribution of the error terms, unlike conventional linear regression models. In this study, we follow the generalized maximum entropy (GME) by Golan, Judge and Perloff (1996). Golan, Judge and Perloff (1996) show that such estimators are more efficient than the ML Tobit estimators. More specifically, it draws inferences from limited or small data using the available observed information from the data to yield a non-uniform distribution with minimal assumptions that is consistent with the observed sample moments (Golan, 2008).

In the GME approach, the entropy of a probability distribution  $\tilde{p}$  is given by

$$H(\tilde{p}) = -\sum (p_i \log p_i)$$

where  $0 \cdot \log 0 \stackrel{\text{def}}{=} 0$ . We seek to maximize this objective function subject to constraints from theory and data (including the constraint  $\sum_i p_i = 1$ ).<sup>13</sup> [See Appendix C for discussion of GME model.]

In addition to the proxy variables and indicators for job quality and care work intensity as well as commute time, we also include controls for selected worker *i* characteristics such as life cycle

<sup>&</sup>lt;sup>12</sup> Based on one of the authors' field interviews with and roundtable presentations by representatives from Seoul Supporting Center for Elderly Care workers, Childcare Workers Chapter of the Korean Confederation of Trade Unions, Seoul LTC Care workers Association, and Korean Domestic Workers' Association, at the International Conference on Empowerment of Care Workers: Issues and Challenges, Seoul National University, Seoul, February 25, 2019. See Moon et al. (2021) for qualitative methodology and survey instruments used in the Care Work and the Economy project's field work in South Korea.

<sup>&</sup>lt;sup>13</sup> The CGE approach seeks to minimize the Kullback-Leibler divergence between  $\tilde{p}$  and another distribution  $\tilde{q}$  from prior knowledge. Maximizing the GME objective function is equivalent to minimizing the CGE objective function with uniform priors for  $\tilde{q}$ . Since we do not have knowledge of a prior distribution, we assume uniform priors, the most conservative assumption.

(age, and age-squared), experience proxied by the number of years since the start of care service employment, years of education, whether the worker resides in a metro area, and if the worker has a spouse. We also control for other job characteristics such as whether performing eldercare or childcare. The basic model is expressed as:

$$\begin{split} Y_{i} &= \beta_{0} + \beta_{1}Age_{i} + \beta_{2}Age_{i}^{2} + \beta_{3}Educ_{i} + \beta_{4}Experience_{i} + \beta_{5}Spouse_{i} + \beta_{6}Metro_{i} \\ &+ \beta_{8}Eldercare_{i} + \beta_{9}ExtraHours_{i} + \beta_{10}NumRecipients_{i} \\ &+ \beta_{11}NeedsConstantWatch_{i} + \beta_{12}CommueTime_{i} + \beta_{13}StableSched_{i} \end{split}$$

+  $\beta_{14}$ FamilyRelation<sub>i</sub> +  $\beta_{15}$ SecureJob<sub>i</sub> +  $\beta_{16}$ InadequateTraining<sub>i</sub> +  $\epsilon_i$ 

where Age<sub>i</sub> is the care worker's age, Educ<sub>i</sub> is the worker's years of education, Experience<sub>i</sub> is the worker's years of experience in providing care work (calculated from the survey question "years since care work first started"), Spouse<sub>i</sub> is an indicator variable for whether the care worker has a spouse, *Metroi* is an indicator variable for whether care work is performed in a metro area, *Eldercare*<sub>i</sub> is an indicator variable for whether the worker is providing eldercare (as opposed to childcare), *ExtraHoursi* is an indicator variable for whether the care worker regularly works extra hours than was originally agreed to (self-reported), *NumRecipients*, refers to the number of care recipients being cared for, NeedsConstantWatchi is an indicator variable for whether the recipient needs to be watched at all times (care worker reports they agree or strongly agree), *CommuteTime* refers to weekly average commute time, *StableSchedi* is an indicator variable for whether the care worker has a stable work schedule (self-reported), FamilyRelation<sub>i</sub> is an indicator variable on whether the care worker reports that it is easy to deal with recipient's family members, *InadequateTraining* dummy indicates if the worker lacks adequate training. *SecureJob<sub>i</sub>* is an indicator variable on whether the worker is a regular (full time) employee, a contract worker with up to 2-year labor contract or a dispatched employee with a signed written agreement and  $\epsilon_i$  is the random error term.

#### 5. Results and Discussion

The results of both Tobit and GME models using the entire sample (both eldercare and childcare workers) are reported in Table 3. The standard errors of the estimates of the latter are smaller since the GME estimators are more efficient. Our results generally hold for both regression analyses, however we focus our discussion on the GME results.

	All Care Workers	
VARIABLES	Tobit	GME
Age	-1.823*	-0.091
	(1.097)	(0.889)
Age-squared	0.0174	0.000
	(0.0112)	(0.009)
Years of education	1.325**	1.417***
	(0.634)	(0.504)
Years since first started care work	0.558	0.085
	(0.349)	(0.245)
Has a spouse)	5.762	1.291
	(3.752)	(2.944)
Metro area worker	-9.574***	-5.777***
	(2.685)	(2.207)
Institution-based worker	-0.957	-2.712
	(2.665)	(2.345)
Eldercare worker	-5.024**	-5.093**
	(2.489)	(2.326)
Worked extra hours	6.464**	5.424**
	(2.651)	(2.147)
Number of recipients under one's care	-1.376	-0.926
	(0.881)	(0.790)
Need to watch care recipient at all times (agree and strongly agree)	5.185**	3.455*
	(2.278)	(1.975)

# Table 3: Tobit and Generalized Maximum Entropy (GME) Regression Estimates:Association between Care Worker's Level of Responsibility for Recipient's Well-being and Working Conditions, by Type of Worker

Daily average commute time (minutes): to and from work	-0.0918**	-0.08**
	(0.0415)	(0.036)
Stable work schedule	10.29***	7.946***
	(2.192)	(2.018)
Easy to deal with recipient's family members	5.465**	4.117**
	(2.339)	(2.065)
Job security proxy <sup>1</sup>	5.329**	3.982
	(2.544)	(2.434)
Received enough training (somewhat or strongly disagree)	-3.520*	-4.242**
	(2.814)	(2.356)
Constant	98.68***	59.848
	(26.75)	(22.594)
Observations	600	600

<sup>1</sup> Dummy variable for worker who is a regular employee, with a signed contract up to 2 years or a dispatched employee with a signed labor contract.

Robust standard errors in parentheses.

\*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Several of the job quality variables are found to be statistically significant. Having a stable work schedule is associated with a 7.9 percentage-point increase at 1 percent level in the reported level of commitment towards the safety and well-being of care recipient, while a lack of adequate training leads to 4.24 percentage-point decline at 5 percent level. The ease in dealing with the recipients' family is associated with a 4.1 percentage-point increase at 5 percent level. This indicates the importance of maintaining a stable work routine that helps avoid sudden and unanticipated changes in the care worker's schedule. Adequate training is also paramount in terms of reducing accidents and building the worker's confidence in dealing with emergencies. It also implies that relationships with the recipient's guardians (parents or children) can affect the care worker's level of commitment. Table 3 results also show that longer commute times are associated with lower reported levels of commitment; that is, an increase in commute time is associated with

an 0.08 percentage-point decrease in respondent's sense of responsibility. These results give support to the Korean care workers' associations' concern about the lack of travel allowance that compel workers to use the cheapest, albeit longer, means of travel to their place of work and about their need for more adequate training.

Interestingly, working extra hours is associated with a 5.4 percentage-point increase in the worker's sense of responsibility, while the need to watch the care recipient at all times is associated with a 3.4 percentage-point increase. We acknowledge that the relationship between the worker's sense of responsibility (the dependent variable) and these care work intensity indicators is bidirectional: on one hand, as workers attempt to meet the intense caregiving needed by the recipient, their sense of commitment also increases. At the same time, workers who care a lot about the recipient may self-select into or stay in positions where the recipient requires being watched at all times.

Table 3 results also indicate that respondent's education has a positive effect i.e., an additional year of schooling is associated with a 1.4 percentage-point increase in the worker's sense of responsibility at 1 percent level. On the other hand, living in a metro area is associated with a 5.8 percentage points decrease at 1 percent level. We hypothesize that this is due to the market density effect in that while there is a larger supply of care workers in the metropolitan areas, there are also more job opportunities in these areas. Hence, it is easier for workers to find other jobs. In smaller cities, towns and the rural areas on the other hand, there are likely to be fewer job opportunities given the lower supply of care centers and care-related institutions, making it harder for care workers to find other employment. Job search outside larger cities is also likely to be based on kinship and community networks. Since communities and neighborhoods in non-metropolitan areas are likely to have more frequent social interactions compared to those in large cities, a care worker could suffer reputational risk if the carer is not dedicated to the recipient's well-being. Performing eldercare is associated with lower sense of responsibility towards the safety and well-being of the recipient compared to childcare. This may be due to the complexity and challenges of caring for older persons.

We next consider the possibility that the relationship between worker's sense of responsibility and working conditions may be different for eldercare workers and childcare workers. We conduct separate Tobit and GME regressions for the childcare and eldercare subsamples and the results are given in Table 4. Note that the standard errors in the subsamples' estimates are larger compared to those for the whole sample in Table 3 due to the smaller sample sizes. We note that working extra hours is positively associated with higher reported level of commitment for both childcare and eldercare workers by 5.1 and 6. 6 percentage points respectively. This also suggests that care workers with strong sense of commitment are willing to work extra hours as needed or requested.

Table 4: Tobit and Generalized Maximum Entropy (GME) Regression Estimates:Association between Worker's Level of Responsibility for Recipient's Well-beingand Working Conditions, by Type of Care Worker

	<b>Childcare Workers</b>		<b>Eldercare Workers</b>	
VARIABLES	Tobit	GME	Tobit	GME
Age	-1.399	0.356	-2.095	1.256
	(1.327)	(1.183)	(2.108)	(2.214)
Age-squared	0.0153	-0.004	0.0187	-0.013
	(0.0141)	(0.013)	(0.0203)	(0.021)
Years of education	1.112	1.647**	1.432*	1.034
	(0.866)	(0.698)	(0.818)	(0.752)
Years since first started care work	0.394	-0.067	0.813	0.656
	(0.386)	(0.311)	(0.503)	(0.435)
Has a spouse	-6.661*	-7.307	8.683**	7.635**
	(3.665)	(4.586)	(4.369)	(3.803)
Metro area worker	-7.192*	-3.198	-9.583***	-7.513**
	(4.105)	(3.284)	(3.168)	(2.962)
Institution-based worker	1.336	-3.141	-1.083	-0.745
	(3.430)	(3.194)	(3.621)	(3.544)
Worked extra hours	5.111*	5.106*	7.159**	6.583**

	(2.881)	(2.792)	(3.622)	(3.317)
Number of recipients under one's care	-2.697**	-1.403	-0.884	-1.161
	(1.146)	(1.218)	(1.108)	(1.094)
Need to watch care recipient at all times (agree and strongly agree)	9.908***	3.597	3.523	3.656
	(2.972)	(2.907)	(2.813)	(2.68)
Daily average commute time (minutes): to and from work	-0.0666	-0.083*	-0.120**	-0.094
	(0.0465)	(0.047)	(0.0579)	(0.059)
Stable work schedule	1.984	3.542	13.12***	11.803***
	(2.860)	(2.934)	(2.669)	(2.795)
Easy to deal with recipient's family member	1.668	1.776	7.730**	8.113***
	(2.617)	(2.756)	(3.079)	(3.11)
Secure job proxy <sup>1</sup>	8.158**	3.454	3.673	3.547
	(3.495)	(3.695)	(3.041)	(3.211)
Received enough training (somewhat or strongly disagree)	-6.034*	-6.78*	-2.694	-2.702
	(3.179)	(3.532)	(3.389)	(3.143)
Constant	100.0***	56.398**	98.59*	16.47
	(34.76)	(27.948)	(53.53)	(58.56)
Observations	300	300	300	300

<sup>1</sup> Dummy variable for worker who is a regular employee, with a signed contract up to 2 years, or a dispatched employee with a signed labor contract.

Robust standard errors in parentheses

\*\*\* p<0.01, \*\* p<0.05, \* p<0.1

The effect of commute time is negatively associated with childcare worker's level of commitment at 10 percent level, but not for eldercare workers. This may be because the majority of childcare workers in Korea work in daycare centers rather than in the homes of recipients, which

can be far from their residence. This makes the commute time a concern for these workers. Half of eldercare workers on the other hand, visit or even live with their clients in the latter's homes. They can select arrangements such that the recipient's house is not too far from the worker's residence or that the cluster of clients live within the same community, which reduces travel time when they move from one client to another on a given day. Lack of adequate training is also associated with a 6.8 percentage-point decline in the childcare worker's level of commitment at 10% level, but not for eldercare workers.

Table 4 results also show that having stable work schedule and ease in dealing with recipient's family member(s) are associated with an increase in 11.8 percentage points and 8.1 percentage points respectively in the eldercare worker's commitment level, but they don't have any statistically significant effect on that of the childcare worker. This may be due to more complicated and heterogeneous nature of eldercare as compared to childcare. Workers caring for older persons are therefore likely to experience difficulties or dilemmas not only in dealing with the recipient's family members but also directly with the elderly. Moreover, the elderly's mental, emotional and physical conditions can change without warning. Since majority of eldercare workers visit or live with their clients at home, maintaining a stable work schedule can be challenging, especially when there are unexpected accidents. <sup>14</sup> Such challenges can lead eventually to greater stress or exhaustion on the part of the care workers, which can affect their level of commitment.

#### 6. Endogeneity Issue

We note in our previous discussion that some of our variables can suffer from endogeneity. In particular, the variables indicating that the worker "usually work extra hours than discussed" and "need to watch care recipients at all times" may suffer from self-selection bias. That is, care workers who are more committed or dedicated might self-select into jobs where they need to always watch the recipient or to work extra hours. Indeed, we find that in the entire sample, care workers who must always watch the care recipient and who work longer hours have a higher sense of responsibility for the recipient on average. This is in contrast to the general expectation that more intense working conditions is associated with lower quality of care (as proxied by the worker's level of commitment variable).

<sup>&</sup>lt;sup>14</sup> For example, lack of bladder control or a fall.

We attempt to correct for this problem by looking at the subsample of care workers who may be randomly assigned by their recipients. We look at a subsample of care workers who are matched to their care recipient by their workplace (this includes both workers who provide care work at a facility and workers who provide home care but work through an institution) to eliminate the self-selection bias.<sup>15</sup> About 22% of home care workers and 55% of institutional care workers are assigned to their recipient by their workplace/care institution.<sup>16</sup>

Another potential source of endogeneity bias is that less dedicated workers might leave if the job is too demanding, leaving the more dedicated workers to work with recipients who need more time or need to be always watched. We correct for this by adding a variable for years of experience in our regression.

We then conduct Tobit and GME regression analyses using this subsample and the results are given in Table 5. We note that "working longer hours than discussed" is no longer associated with a higher sense of responsibility to the recipient. However, the "need to watch care recipient at all times" remains statistically significant, suggesting that having to watch the care recipient at all times increases the sense of responsibility the care worker feels towards the recipient.<sup>17</sup>

<sup>&</sup>lt;sup>15</sup> We note that these variables can suffer from other forms of endogeneity. For example, care workers who feel less committed to might attrite from care work at higher rates when working conditions are intense, leaving only the more committed workers in our sample (survivor bias). In addition, workplaces might try to match more dedicated workers with more difficult cases, in which case, our assumption that "workplace assignment" would serve as a randomizing mechanism no longer holds.

<sup>&</sup>lt;sup>16</sup> See Appendix D Table D1.

<sup>&</sup>lt;sup>17</sup> For the subsample of institutionally assigned workers, we also examine the group mean of the reported level of responsibility for the bottom 20% of workers by experience (those with 0-2 years of experience) and the top 20% of workers by experience (those with 8-30 years of experience). The mean level of responsibility for those with 0-2 years of experience is 72.9% and for those with 8-20 years of experience is 72.2%. The differences in means are not statistically significant. Note that the cases whereby spending more time or watching the care recipient causes the care worker to feel more responsible for the recipient is not endogenous. We believe that this is the causal effect of spending more time with the recipient.

	<b>Childcare Workers</b>		<b>Eldercare Workers</b>	
VARIABLES	Tobit	GME	Tobit	GME
Age	-2.630	2.116	-11.37***	2.425
	(1.659)	(4.626)	(2.675)	(5.015)
Age-squared	0.0347*	-0.031	0.108***	-0.027
	(0.0196)	(0.054)	(0.0270)	(0.048)
Years of education	3.904***	-1.263	-0.849	2.777
	(1.279)	(2.651)	(1.264)	(2.037)
Years since first started care work	-0.248	0.318	0.612	0.587
	(0.483)	(1.02)	(0.648)	(0.977)
Has a spouse	-5.027	-0.633	25.78***	-1.306
	(5.839)	(13.537)	(5.717)	(9.191)
Metro area worker	-13.51**	-10.12	-3.497	-7.909
	(5.952)	(13.1)	(5.089)	(7.239)
Worked 40 hours or more	0.444	-2.494	8.762	-9.501
	(8.109)	(20.005)	(6.842)	(10.636)
				-6.248
Worked extra hours	3.097*	2.525	-0.415	(8.91)
	(4.313)	(10.288)	(5.769)	-4.04
Number of recipients under one's care	-1.226	-1.979	-3.433***	(2.904)
	(2.184)	(4.705)	(1.709)	3.048
Need to watch care recipient at all times	1/1 Q1***	0 847	13 20***	(7 062)
(agree and strongly agree)	14.01	(12.205)	(4.092)	(7.005)
Daily avarage commute time (minutes): to	(5.662)	(12.265)	(4.983)	0.162
and from work	-0.0651	0.075	-0.0630	(0.134)

# Table 5 Tobit and GME Regression Estimates for institutionally assigned subsample

	(0.0825)	(0.176)	(0.0662)	-40.84
Stable work schedule	2.719	4.993	16.84***	(6.908)
	(4.395)	(11.435)	(3.968)	-4.818
Easy to deal with recipient's family member	7.098	-2.755	9.345***	(7173)
	(4.606)	(10.466)	(4.437)	-9.485
Secure job proxy <sup>1</sup>	1.779**	-7.603	-0.136	(8.223)
	(4.731)	(14.089)	(5.477)	-11.051
Received enough training (somewhat or	17 46***	7 6 4 2		(7.77)
strongly disagree)	-17.46****	7.643	-5.506	(7.272)
	(5.495)	(14.785)	(5.319)	-0.368
Constant	72.92*	13.58	345.8***	(139.753)
	(39.21)	(93.294)	(68.50)	-9.501
Observations	200	200	250	250

<sup>1</sup> Dummy variable for worker who is a regular employee, with a signed contract up to 2 years, or a dispatched employee with a signed labor contract.

Robust standard errors in parentheses

\*\*\* p<0.01, \*\* p<0.05, \* p<0.1

We note a third potentially endogenous variable in our analysis: ease in dealing with the care recipient's family members. Care workers with higher level of commitment could be better treated by family members, compared to those with lower level of commitment. Thus, the quality of care provided by the care worker might be causing the difficulty or easiness of dealing with family members, rather than the other way around. On the other hand, family members might try to take advantage of more dedicated care workers, saddling them with more of the care responsibility, and souring the relationship between care worker and family members. Thus, we do not know the direction of bias in this study, given data limitations. Future research can explore this relationship and can help shed light on this aspect of care work.

#### 7. Conclusion

Korea's puzzling situation of continued heavy reliance on family care for the frail older persons and young children, despite the ample supply of paid care services and workers has brought attention to the quality of care services. This issue is complex and yet critical in our understanding of the conundrum faced by Korea and other countries and in developing an integrated policy agenda aimed at the provision of affordable, quality childcare and eldercare. While a range of factors influence the performance and delivery of care provided by workers - from the nature of care infrastructure, level of employee training, to access to complementary resources, this paper focuses on a crucial factor i.e., the care worker's sense of responsibility for the well-being of the recipient. In other words, carers should care about those they care for as the latter's well-being depends upon their emotional support and effort that they provide.

Research on care have pointed out that care work, whether paid or unpaid, has motivational dimensions, in addition to practical dimensions. A worker's commitment to their care recipient's well-being is not static however, it can change and can vary over time depending on circumstances. We argue in this paper that working conditions e.g., job quality, work intensity, and relationship with recipient's family are related with a worker's level of commitment. In this sense, working conditions can affect the quality of care.

Using the 2018 Care Work Economy (CWE-GAM) survey of 600 childcare and eldercare workers in South Korea, we assess the relationship between commitment levels among these workers using tobit and GME tests. Our results suggest that longer commute time, a lack of adequate training, particularly for childcare workers, are associated with lower levels of commitment. Our results also suggest that having stable work schedule and ease in dealing with family members, particularly for eldercare workers, are correlated with higher levels of commitment. That said, our findings are only tentative since our analysis admittedly has omitted variable bias due to data limitations.<sup>18</sup>

Our results show that the expressed level of commitment is not impervious to the working conditions that the care worker faces. Policies that promote decent working conditions such as the provision of stable work schedule and adequate training, travel allowance for care workers to

<sup>&</sup>lt;sup>18</sup> For example, paid care worker's emotional and relational skills, which are difficult to measure, are not taken into account. We are also unable to take hourly wage into consideration as this was not collected in the survey.

reduce commute time, as well as guidelines that promote a healthy relationship between recipientfamily and hired-care worker are critical in improving the quality of care services. Our study findings also show that many childcare and eldercare workers in Korea work extra hours and, in some cases, intensely as their recipient needs to be watched at all times. The issues of quality care services and the working condition of care workers are intricately connected. Our results are consistent with findings from the experimental investigation of care provisioning by Banuri et al. (2019), which show that increasing the effectiveness of care workers may significantly impact the quality of work provided. Improving the working conditions of care workers, in addition to providing care workers with adequate training will likely yield higher quality of care services. With the growing need for care and families' concern for quality care, it is time for policymakers to seriously consider the nature of care workers' employment conditions and need for adequate training in efforts to promote quality care.

#### REFERENCES

- 2018 South Korea Paid Careworker Survey- Eldercare. Care Work and The Economy Project Field Work Data, Center for Transnational Migration and Social Inclusion, Seoul National University.
- 2018 South Korea Paid Careworker Survey- Childcare. Care Work and The Economy Project Field Work Data, Center for Transnational Migration and Social Inclusion, Seoul National University.
- Anna Louie, S., 2020, 'Women's Work' Can No Longer Be Taken for Granted. New York Times Company, New York.
- Araujo, M.C., Dormal, M., Schady, N., 2019. Childcare Quality and Child Development. Journal of Human Resources, 54(3), 656-682.
- Banuri's, S., De Oliveira, A.C.M., Eckel, C.C., 2019. Care Provision: An Experimental Investigation. Journal of Economic Behavior & Organization, 157, 615-630.
- Brekke, K.A., Nyborg, K., 2010. Selfish Bakers, Caring Nurses? A Model of Work Motivation. Journal of Economic Behavior & Organization, 75(3), 377-394.
- Burchinal, M., Howes, C., Kontos, S., 2002. Structural Predictors of Child Care Quality in Child Care Homes. Early Childhood Research Quarterly, 17(1), 87-105.
- Burke, M.J., Sarpy, S.A., Smith-Crowe, K., Chan-Serafin, S., Salvador, R.O., Islam, G., 2006. Relative Effectiveness of Worker Safety and Health Training Methods. American Journal of Public Health, 96(2), 315-324.
- Cha, S.E., Kang, E., Floro, M., Arslan, S., Alam, A., 2022, Care Conundrum" in South Korea Unmet Demand for Childcare and Eldercare. Care Work and The Economy Working Paper Series. American University, Washington, DC.
- Choi, I.-H., Kim, J.-H., 2013. Characteristics of Certified Family Care Providers and Informal Caregivers, and Predictors of Their Caregiving Burden. Family and Culture, 25(3), 159-185.
- Choi, I.-H., Kim, Y.-R., Yeom, J., Kim, S., 2012. The Profile of Family Caregiving as Provided by Female Older Adults in Korea. Gender Rev, 27, 74-77.
- Choi, I.H., Kim, Y.R., Lee, A.R., Park, S., 2014. Family Caregiving in An Aging Society: Evaluation of The Current Family Support System and Policy Implications. Seoul, South Korea: Korean Women's Development Institute.
- Chung, K., Oh, Y., Lee, Y., Oh, M., Kang, E., Kim, K., Hwang, N., Kim, S., Lee. S., Lee. S., Hong, S., 2017, Survey of Living Conditions and Welfare Needs of Korean Older Persons [Policy Report]. Ministry of Health and Welfare Sejong.
- Colwell, N., Gordon, R.A., Fujimoto, K., Kaestner, R., Korenman, S., 2013. New Evidence on The Validity of The Arnett Caregiver Interaction Scale: Results from The Early Childhood Longitudinal Study-Birth Cohort. Early Childhood Research Quarterly, 28(2), 218-233.
- Coverman, S., 1987, "Households, Employment, and Gender: A Social, Economic and Demographic View" By Paula England and George Farkas (Book Review). Sage Publications, Newbury Park, Calif, Pp. 476.

- Early, D.M., Sideris, J., Neitzel, J., Laforett, D.R., Nehler, C.G., 2018. Factor Structure and Validity of The Early Childhood Environment Rating Scale Third Edition (ECERS-3). Early Childhood Research Quarterly, 44, 242-256.
- Eaton, S.C., 2005. Eldercare in the United States: Inadequate, Inequitable, But Not a Lost Cause. Feminist Economics, 11(2), 37-51.
- England, P., 2005. Emerging Theories of Care Work. Annual Review of Sociology, 31(1), 381-399.
- England, P., Farkas, G., 1986, Households, Employment, and Gender: A Social, Economic, and Demographic View / Paula England and George Farkas. Aldine Pub. Company, New York.
- Folbre, N., 2006. Demanding Quality: Worker/Consumer Coalitions and "High Road" Strategies in The Care Sector. Politics & Society, 34(1), 11-32.
- Folbre, N., 2008. When A Commodity Is Not Exactly a Commodity. Science, 319(5871), 1769-1770.
- Folbre, N., Nelson, J.A., 2000. For Love or Money -- or Both? The Journal of Economic Perspectives, 14(4), 123-140.
- Folbre, N., Weisskopf, T., 1998. Did Father Know Best? Families, Markets, and The Supply of Caring Labor. Economics, Values and Organization, 171-205.
- Forder, J., Allan, S., 2014. The Impact of Competition on Quality and Prices in The English Care Homes Market. Journal of Health Economics, 34(100), 73-83.
- Gallup, K., 2018, 2018 South Korea Paid Care Worker Survey Eldercare. Gallup 2018-166-007-01. Gallup Korea, Graduate School of International Studies, Seoul National University.
- Gallup, K., 2018, 2018 South Korea Paid Care Worker Survey Childcare. Gallup 2018-166-007-01. Gallup Korea, Graduate School of International Studies, Seoul National University.
- Gershon, R.R., Vlahov, D., Felknor, S.A., Vesley, D., Johnson, P.C., Delclos, G.L., Murphy, L.R., 1995. Compliance With Universal Precautions Among Health Care Workers at Three Regional Hospitals. American Journal of Infection Control, 23(4), 225-236.
- Golan, A., 2008. Information and Entropy Econometrics—A Review and Synthesis. Foundations and Trends® in Econometrics, 2(1–2), 1-145.
- Golan, A., Judge, G.G., Miller, D., 1996, Maximum Entropy Econometrics: Robust Estimation with Limited Data / Amos Golan, George Judge, Douglas Miller. Wiley, Chichester [England].
- Harding Weaver, R., 2002. Predictors of Quality and Commitment in Family Child Care: Provider Education, Personal Resources, and Support. Early Education and Development, 13(3), 265-282.
- Hensly, C., King, E., Jun, J., 2021, The Toll and Reward of Family Caregiving: Eldercare in South Korea. In: A. University (Ed.). Care Work and The Economy Working Paper Series. American University, Program on Gender Analysis in Economics, Washington, DC.
- Himmelweit, S., 1995. The Discovery of "Unpaid Work": The Social Consequences of The Expansion of "Work". Feminist Economics, 1(2), 1-19.
- Himmelweit, S., Land, H., 2010. Change, Choice and Cash in Social Care Policies: Some Lessons from Comparing Childcare and Elder Care, Open Discussion Papers in Economics.

- Hong, S.Y., Seo, J.H., 2012. The Study on Commodification of Filial Piety Through Long Term Care Service. Korean Journal of Gerontological Social Welfare (57), 51-84.
- Jakobsson, N., Kotsadam, A., Syse, A., Øien, H., 2016. Gender Bias in Public Long-Term Care? A Survey Experiment Among Care Managers. Journal of Economic Behavior & Organization, 131, 126-138.
- Joling, K.J., Van Eenoo, L., Vetrano, D.L., Smaardijk, V.R., Declercq, A., Onder, G., Van Hout, H.P.J., Van Der Roest, H.G., 2018. Quality Indicators for Community Care for Older People: A Systematic Review. Plos One, 13(1), E0190298-E0190298.
- Jones, G.W., 2019. Ultra-Low Fertility in East Asia: Policy Responses and Challenges. Asian Population Studies, 15(2), 131-149.
- Jun, J., Eun, K.S., Jang, E., Cha, S.E., Kang, E., 2021, Quantitative Methodology and Survey Instruments and Unpaid Care Work. American University.
- Kang, E., Eun, K.S., Jun, J., Cha, S.E., Moon, H., 2021, Care Arrangement and Activities in South Korea: An Analysis of the 2018 Care Work Family Survey on Childcare and Eldercare. Care Work and The Economy Working Paper Series. American University, Washington, DC.
- KICCE, 2018, National Early Education Statistics Report. Korean Institute of Child Care and Education (KICCE).
- Kilaberia, T.R., 2020. Organizational Commitment Among Residential Care Workers. Journal of Aging Studies, 55.
- Kim, D.R., Kim, K.H., Lee, S.Y., Han, S.H., Lee, G.R., 2021. Influencing Factors Affecting Caregivers' Service Quality for The Elderly with Dementia: Focusing on Dementia Knowledge, Work Value, and Job Performance Confidence. Journal of Korean Academy of Community Health Nursing, 32(3), 233-242.
- Kim, E.-Y., Yeom, H.-E., 2016. Influence of Home Care Services on Caregivers' Burden and Satisfaction. Journal of Clinical Nursing, 25(11-12), 1683-1692.
- Kim, J., De Bellis, A.M., Xiao, L.D., 2018. The Experience of Paid Family-Care Workers of People with Dementia in South Korea. Asian Nursing Research, 12(1), 34-41.
- Kim, S., 2020. Latent Profiles Analysis of Risk and Protective Factors Predicting Child Abuse in Daycare Centers. Early Childhood Education Research & Review, 24(1), 41-66.
- Kim, S., Jeong, M., 2006. A Study on Working Mother's and Grandmother's Perceptions of Childcare. The Journal of Korea Open Association for Early Childhood Education, 11(5), 67-96.
- Kim, Y.K., 2019, The Double Care Burden and Policy Implications for Middle-Aged and Older Heads. Health and Welfare Forum, Pp. 74-91.
- Kondo, K., 2015. Evaluation of Effectiveness, Quality and Inequalities in Health, Medical and Long-Term Care-Achievements and Challenges. Public Money Manag, 11(5), 685-718.
- Kwon, H.J., Hong, G.J., 2017. The Working Conditions for Care Workers and Care Quality in Long-Term Care Services. Korean Journal of Social Welfare, 69(1), 33-57.

- Lee, J., 2018, Support for The After-School Care for Elementary School Children. Korea Institute of Child Care and Education.
- Lee, J., Kim, E.Y., 2019. A Meta-Analysis on Potential Factors Affecting Young Child Abuse. The Journal of Korea Open Association for Early Childhood Education, 24(3), 97-119.
- Lee, S., Cho, M., Lee, J., 2015, A Study on Preliminary Research and Public Policy for Various Family Care Services. Seoul Foundation of Women & Family.
- Lee, Y.K., 2021, The 2021 Outlook for Population Policy. Health and Welfare Policy Forum. Korea Institute for Health and Social Affairs, Pp. 35-45.
- Mashburn, A.J., Pianta, R.C., Hamre, B.K., Downer, J.T., Barbarin, O.A., Bryant, D., Burchinal, M., Early, D.M., Howes, C., 2008. Measures of Classroom Quality in Prekindergarten and Children's Development of Academic, Language, and Social Skills. Child Development, 79(3), 732-749.
- Meagher, G., 2007. The Challenge of The Care Workforce: Recent Trends and Emerging Problems. The Australian Journal of Social Issues, 42(2), 151-167.
- Meyer, J.P., Maltin, E.R., 2010. Employee Commitment and Well-Being: A Critical Review, Theoretical Framework and Research Agenda. Journal of Vocational Behavior, 77(2), 323-337.
- MOGEF, 2012, Project Mannual on Ai-Dolbom Support. Ministry of Gender Equality and Family.
- MOGEF, 2012, Project Manual on Cooperative Childcare Program. Ministry of Gender Equality and Family.
- MOHW, 2010, Program Guideline for Community Child Care Center. In: M.O.H.A. Welfare (Ed.), Sejong, Korea.
- MOHW, 2017, 2017 Eldercare Facility Statistics. Ministry of Health and Welfare (MHW), Seoul, Korea.
- MOHW, 2018, 2017 Day Care Centre Statistics. Ministry of Health and Welfare (MHW), Seoul, Korea.
- MOHW, 2019, Project Manual on Customized Elderly Care Service. In: M.O.H.A. Welfare (Ed.), Sejong, Korea.
- MOHW, 2020, 2019 Child Care Statistics. Ministry of Health and Welfare (MHW), Sejong, Korea.
- Moon, H., Cha, S.E., 2020. Family Caregivers' Elderly Care: Understanding Their Hard Time and Care Burden. Family and Culture, 32(1), 102-131.
- Moon, H., Cha, S.E., Eun, K.S., Kang, E.H., Peng, I., 2021, The Qualitative Methodology and Survey Instruments: South Korea Field Work Care Work and the Economy Methodological Report American University, Program on Gender Analysis in Economics, Washington, DC.
- Nelson, J.A., 1999. of Markets and Martyrs: Is It OK To Pay Well for Care? Feminist Economics, 5(3), 43-59.
- Nelson, J.A., 2010. Getting Past "Rational Man/Emotional Woman": Comments on Research Programs in Happiness Economics and Interpersonal Relations. International Review of Economics, 57(2), 233-253.

- NHIC, 2017, 2017 Long Term Care Insurance Statistical Yearbook. National Health Insurance Corporation (NHIC), Seoul, Korea.
- NHIC, 2019, 2019 Long Term Care Insurance Statistical Yearbook. National Health Insurance Corporation (NHIC), Seoul, Korea.
- NHIS, 2016, Long-Term Care Insurance Statistical Yearbook 2013-2016. National Health Insurance Service, Wonju, Korea.
- Nicol, R., Petrina Sweeney, M., Mchugh, S., Bagg, J., 2005. Effectiveness of Health Care Worker Training on The Oral Health of Elderly Residents of Nursing Homes. Community Dentistry and Oral Epidemiology, 33(2), 115-124.
- NSO, 2017, 2017 Day Care Centre Statistics Yearbook. National Statistics Office, Seoul.
- Ochiai, E., 2009. Care Diamonds and Welfare Regimes in East and South-East Asian Societies: Bridging Family and Welfare Sociology. International Journal of Japanese Sociology: IJJS, 18(1), 60-78.
- OECD, 2019, Gender Wage Gap.
- OECD, 2020, OECD Economic Surveys: Korea 2020. OECD Publishing, Paris.
- Park, K.J., Hwang, O.K., Moon, H.J., 2013. Childcare Policies in Korea. Korean Journal of Childcare and Education, 9(5), 513-538.
- Peng, I., Cha, S.E., Moon, H., 2020, An Overview of Care Policies and The Status of Care Workers in South Korea. Care Work and The Economy Working Paper Series. American University, Washington, DC.
- Rios-Zertuche, D., Zúñiga-Brenes, P., Palmisano, E., Hernández, B., Schaefer, A., Johanns, C.K., Gonzalez-Marmol, A., Mokdad, A.H., Iriarte, E., 2019. Methods To Measure Quality of Care and Quality Indicators Through Health Facility Surveys in Low- and Middle-Income Countries. International Journal for Quality in Health Care, 31(3), 183-190.
- Rubio-Codina, M., Dormal, M., Araujo, M.C., 2019. Observing Home-Visiting Quality at Scale with The Home Visit Rating Scales and A Supervisor Checklist in Peru. Infant Mental Health Journal, 40(3), 343-362.
- Ruffini, K., 2021, Worker Earnings, Service Quality, and Firm Profitability: Evidence from Nursing Homes and Minimum Wage Reforms. Social Science Research Network.
- Shewark, E.A., Zinsser, K.M., Denham, S.A., 2018. Teachers' Perspectives on The Consequences of Managing Classroom Climate. Child & Youth Care Forum, 47(6), 787-802.
- Shin, J.H., Hyun, T.K., 2015. Nurse Staffing and Quality of Care of Nursing Home Residents in Korea. Journal of Nursing Scholarship, 47(6), 555-564.
- Siciliani, L., Rune Straume, O., Cellini, R., 2013. Quality Competition with Motivated Providers and Sluggish Demand. Journal of Economic Dynamics & Control, 37(10), 2041-2061.
- Song, D., 2014. A Study on Double-Care and Multiplicity of Caring Experiences Among Women Aged 30s to 40s in Korea. Korean Journal of Social Welfare, 66(3), 209-230.

- Song, D., 2016. A Study of The Double Burden and Everydayness of Caring from Caregiver Perspective. PNU Journals of Women's Studies, 262(2), 127-161.
- SPASS, 2020, Project Manual on Social Service One. Seoul Public Agency for Social Service (SPASS), Seoul, Korea.
- Steinberg, R.J., 1999. Emotional Labor in Job Evaluation: Redesigning Compensation Practices. The Annals of The American Academy of Political and Social Science, 561(1), 143-157.
- Suh, J.Y., 2020, Estimating the Paid Care Sector in South Korea. Care Work and The Economy Working Paper Series. American University, Program on Gender Analysis in Economics, Washington, DC.
- Suh, J.Y., 2021, Estimating the Unpaid Care Sector in South Korea. Care Work and The Economy Working Paper Series. American University, Program on Gender Analysis in Economics, Washington, DC.
- Sung, S., 2018. Gender, Work and Care in Policy and Practice: Working Mothers' Experience of Intergenerational Exchange of Care in South Korea. Critical Social Policy, 38(3), 589-608.
- Teng, C.-I., Dai, Y.-T., Lotus Shyu, Y.-I., Wong, M.-K., Chu, T.-L., Tsai, Y.-H., 2009. Professional Commitment, Patient Safety, and Patient-Perceived Care Quality. Journal of Nursing Scholarship, 41(3), 301-309.
- Tronto, J.C., 1987. Beyond Gender Difference to A Theory of Care. Signs: Journal of Women in Culture and Society, 12(4), 644-663.
- Waerness, K., 1984. The Rationality of Caring. Economic and Industrial Democracy, 5(2), 185-211.
- Westerberg, K., Pienaar, J., Nordin, M., Romeo, M., Yepes-Baldó, M., 2021. Organizational Change and Commitment: Effects on Well-Being, Turnover Intent and Quality of Care in Spanish and Swedish Eldercare. Economic and Industrial Democracy, 42(4), 899-916.

# APPENDICES

## Appendix A. Current Major Eldercare and Eldercare Programs in Korea

#### A1: List of Major Eldercare Programs in Korea

Long-Term Care (LTC) Insurance Program	This program was established in 2008 in response to the rising costs of eldercare and the growing burden of families providing care to older persons. Persons aged 65 years and older or those who suffer from geriatric illness are eligible. The program provides facility-based care such as nursing home, elderly daycare center, short-term respite care, as well as home-based care such as home-visit care, home-visit bathing, home-visit nursing, short-term housework service, and welfare equipment services. The National Health Insurance Corporation (NHIC) administers and supervises the LTC program, supervised by the Ministry of Health and Welfare.
Customized Elderly Community Care System	The customized elderly community care system was implemented in 2020 in Korea by the Ministry of Health and Welfare. It consolidates previous elder care programs including basic elder care service, the comprehensive elder care service, programs supporting older persons living alone, short-term housework service, and community care. The local governments consign the program to social care centers or non-profit organizations in providing the service. The customized elderly community care system is operated separately from the LTC program. LTC beneficiaries are not eligible for receiving this service.
Social Service One Program	Social Service One is a public foundation established in 2019 in Korea in response to strengthening social care. Care provided by small private businesses raised concerns in Korea due to their low quality of care and mismanagement. Social Service One aims to providing quality care and quality jobs in the care economy through direct employment of care workers and direct operations of elder care services. A consigned management of daycare centers for children is also part of their program. SSO services are currently available in four areas of Korea: Seoul, Gyeonggi-do, Daegu, and Gyeongsangnam-do and will be expanded to 17 additional areas. So far, older persons, children, and the disabled are the main beneficiaries. SSO plans to include individuals with different needs in its service provisioning.

Sources:

1) National Health Insurance Corporation. 2019 National Long-Term Care Insurance Statistics Yearbook.

http://www.mohw.go.kr/react/jb/sjb030301vw.jsp.

3) Seoul Public Agency for Social Service. 2020. Project Manual on Social Service One. https://seoul.pass.or.kr/sub0604/file\_down/id/591

A2: List of Current Major Childcare Programs in Korea

<sup>2)</sup> Ministry of Health and Welfare. Project Manual on Customized Elderly Care Service.

Universal	In 2013, the Korean government expanded its financial support to all
Childcare and	children aged 0-5 who are registered in daycare centers or in kindergarten
Kindergarten	through tuition subsidy. This has led to high enrollment rates of children in
Tuition Subsidy	daycare centers or kindergarten.
Program	
Public	The <i>Ai-dolbomi</i> service program is a public, home-visit babysitting service
Babysitting	implemented in 2007 that provides short-term, temporary care to children
Service (Ai-	aged from 3 months to 12 years old. It was a response to the increase in
dolbomi)	dual-earner households that led to a growing demand for such services,
Program	which is not covered by existing care service programs in Korea. Over
	70,000 households who used the service in 2019.
Community	This program provides care to children from low-income families and from
Child Care	dual-earning parent households. With the building and operation of
Center Program	community care centers after the amendment of the Child Welfare Act in
(Jiyeok-adong	2004, the program provides more than 8 hours of care that include various
Center Program)	activities for children as well as meals and snacks. The center is open
	during school vacations and is freely available to children from low-
	income families in need of social protection, and to other children for a fee,
	if there is vacancy.
Cooperative	This program was established in 2018 in response to the growing need for
Childcare	care sharing among parents. It is rooted in <i>Poom-asi</i> , a traditional way of
Program	taking care of children in neighborhoods in Korean society. This
(Gongdong-	community-sharing childcare is done at neighborhood centers where
yooka-	parents take turns in providing care. The program offers various activities
nanumteo)	to children and provides opportunities to meet neighbors as well as share
	care knowledge.

Sources:

1) Childcare Policies in Korea. (Park et al., 2013).

2) Ministry of Gender Equality and Family. 2012. Project Manual on Ai-dolbom Support.

http://www.mogef.go.kr/kor/skin/doc.html?fn=eff1086d02244697bba022baab4fe775.pdf&rs=/rsfiles/202203/

3) Ministry of Health and Welfare. 2010. Program Guideline for Community Child Care Center.

4) Ministry of Gender Equality and Family. 2012. Project Manual on Cooperative Childcare Program. http://www.mogef.go.kr/kor/skin/doc.html?fn=21198.hwp&rs=/rsfiles/202203/

#### **Appendix B: Construction of Survey Weights**

The eldercare and childcare worker survey data collection in Korea for the Care Work and the Economy (CWE-GAM) Project was performed in 2018 using a purposive sampling design (Jun et al., 2021). The 600 samples were evenly split between eldercare and childcare workers. For eldercare workers, of the 300 workers surveyed, 150 samples were allocated to institutional workers, 100 to inhome care workers and 50 to informal workers. These samples were further stratified by region namely, Seoul Metro, Chungcheong, Honam, Gyungbuk, and Gyungnam. For childcare workers 100 samples were allocated to institutional workers (50 to public daycare centers, 50 to private daycare centers), 100 samples were allocated to in-home care workers and 100 samples were allocated to informal workers.

	Elo	Childcare workers					
	Institution	In-home	Informal	Institution		In-home	Informal
				Public	Private		
Seoul Metro	80	42	10	25	27	20	20
Chungcheong	20	11	10	6	6	20	20
Honam	18	23	10	8	5	20	20
Guyngbuk	16	10	10	5	4	20	20
Guyngnam	16	14	10	6	8	20	20
Total	150	100	50	50	50	100	100

#### Table B1. Sample Allocation

We weighted the purposive sample used in the paper to make it representative of the eldercare and childcare worker population in Korea by calculating the inverse sampling probability weight for each observation. For institutional eldercare workers, the relevant subpopulation was the number of institutional workers in each region published in the 2017 Eldercare Facility Statistics (Ministry of Health and Welfare, 2017a). For in-home eldercare workers, the relevant subpopulation was the number of inhome care workers in each region as published in the 2017 Long-Term Care Insurance Statistical Yearbook. For childcare workers, the relevant subpopulation was the number of care workers (excluding administrative staff and instructors) for each type of institution (private facility, private in-home or public) in the region, as published in the 2017 Day Care Centre Statistics (National Statistics Office, 2017).

The number and distribution of informal care workers across Korea is unknown, so we use the estimates on the number of informal childcare and eldercare workers using the method in Suh (2020) paid care sector in Korea study. We assumed that the distribution of informal care workers among childcare and eldercare worker subpopulation follows the same pattern as that of formal care workers. That is, about a third were employed in childcare while the rest were employed in eldercare. We next assumed that the regional distribution of workers follows the regional GDP share. The relevant subpopulation for informal care workers is the estimated number of informal workers in each region for each type of care work (childcare or eldercare).<sup>19</sup>

The sampling probability  $p_i$  for an observation in subpopulation *i* is simply the number of samples allocated to the subpopulation  $n_i$  divided by the size of the subpopulation  $N_i$ .

$$p_i = \frac{n_i}{N_i} \tag{c1-1}$$

The inverse sampling probability weight is  $1/p_i$ .

<sup>&</sup>lt;sup>19</sup> For example, to obtain the survey weight for informal childcare workers in Chungcheong Area: we use the total number of informal childcare workers: 27,500; and Chungcheong's share of Korean GDP: 13.45%; to get estimated number of informal childcare workers: 3,700 = 27,500\*13.45%. We then divide this by the number of informal childcare workers surveyed in Chungcheong (20) to obtain the survey weight 2700/20 = 185.

	Institutional Workers			In-Home Workers			Informal Workers		
Region	Total Workers <i>N<sub>i</sub></i>	Workers Surveyed n <sub>i</sub>	Weight p <sub>i</sub>	Total Workers	Workers Surveyed	Weight	Total Workers	Workers surveyed	Weight
Seoul/Metro	47,688	80	596.10	10,955	42	260.83	19,372	27	717.49
Chungcheong Area	11,969	20	598.45	2,736	11	248.73	5,010	6	834.97
Honam Area	11,206	18	622.56	6,005	23	261.09	3,383	5	676.62
Gyungbuk Area	9,707	16	606.69	2,409	10	240.90	3,376	4	843.93
Gyungnam Area	9,373	16	585.81	3,739	14	267.07	5,918	8	739.71
Total	89,943	150		25,844	100		37,055	50	

# Table B2. Survey Weights for Eldercare Workers

## Table B3. Survey weights for childcare institutional care workers

	Institutional									
	Public	Non-Profit	Workers Surveyed	Weights	Private	Workers Surveyed	Weights			
Seoul/Metropolitan	2,179	218	25	95.88	6,988	27	258.81			
Chungcheong Area	195	277	6	78.67	1,614	6	269.00			
Honam Area	191	426	8	77.13	1,326	5	265.20			
Gyungbuk Area	212	203	5	83.00	1,532	4	383.00			
Gyungnam Area	351	193	6	90.67	2,352	8	294.00			
Total	3,128	1317	50		13,812	50				

		In-Home		Informal			
	In-Home	Workers Surveyed	Weights	Informal	Workers Surveyed	Weights	
Seoul/Metropolitan	10,998	20	549.90	9,382	20	469.08	
Chungcheong Area	2,591	20	129.55	2,426	20	121.31	
Honam Area	1,767	20	88.35	1,638	20	81.92	
Gyungbuk Area	1,424	20	71.20	1,635	20	81.74	
Gyungnam Area	2,741	20	137.05	2,866	20	143.29	
Total	19,521	100		17,945	100		

# Table B4. Survey wights for childcare in-home and informal care workers

#### Appendix C. Discussion of the Generalized Maximum Entropy (GME) Model

In the case of the GME model, we assume that the  $\beta$  are discrete random variables drawn from a support space  $\mathcal{L} \subset \mathcal{R}^k$  where k is the number of parameters in the problem. Then  $\beta$  maybe expressed as

$$\beta = \begin{bmatrix} z_1 & 0 & . & 0 \\ . & z_2 & . & 0 \\ . & . & . & . \\ 0 & 0 & . & z_k \end{bmatrix} \begin{bmatrix} p_1 \\ p_2 \\ . \\ p_k \end{bmatrix}$$

Similarly, we assume that the errors from the model are being drawn from some discrete bounded distribution. Thus, the error distribution maybe written as

$$e = Vw = \begin{bmatrix} v_1 & 0 & . & 0 \\ . & v_2 & . & 0 \\ . & . & . & . \\ 0 & 0 & . & v_k \end{bmatrix} \begin{bmatrix} w_1 \\ w_2 \\ . \\ . \\ w_k \end{bmatrix}$$

Where *w* are the probability weights associated with each outcome. Then our objective function becomes (bold-faced variables indicate vectors or matrices)

$$\max_{p,w_1,w_2,w_3} - \boldsymbol{p}^T \log \boldsymbol{p} - \boldsymbol{w}_1^T \log \boldsymbol{w}_1 - \boldsymbol{w}_2^T \log \boldsymbol{w}_2^T - \boldsymbol{w}_3^T \log \boldsymbol{w}_3$$

Subject to the constraints

$$y_1 = X_1 Z_p + V_1 w_1$$
$$0 = \mu_1 \le X_2 Z_p + V_2 w_2$$
$$1 = \mu_2 \ge X_3 Z_p + V_3 w_3$$

and the adding up constraints described in Golan, Judge, Perloff (1996) eq. 4.6 - 4.8. Note that our responses are bound on both sides, so we have an additional data constraint and adding up constraint.

The estimation procedure requires the researcher to make several choices. For the support space Z, we choose

$$\boldsymbol{Z} = \begin{bmatrix} -100 & -50 & 0 & 50 & 100 \\ -100 & -50 & 0 & 50 & 100 \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ -100 & -50 & 0 & 50 & 100 \end{bmatrix}$$

Where **Z** is of dimension 20 × 5. Golan, Judge and Perloff (1996) show that as long as  $Z_{1k} \le \beta_k \le Z_{Hk}$ , the estimates are not very sensitive to the specification of the support space. (In our case, H =

5 and we assume the  $\beta_k$  are bound between [100, 100] For the error supports, we use the 3-sigma rule for  $v_1$  and choose uniform errors between [10, 10] for  $v_2$  and  $v_3$ . That is:

$$V_2 = V_3 \begin{bmatrix} -10\\0\\10 \end{bmatrix}$$

We test with alternative specifications of  $V_2$  and  $V_3$  and note they do not significantly change the result.